RESEARCH Open Access



The context of emergency contraception use among young unmarried women in Accra, Ghana: a qualitative study

Slawa Rokicki¹ and Sonja Merten^{2,3*}

Abstract

Background: Over the past decade, awareness and use of emergency contraceptive pills (ECPs) among young women has rapidly increased in Ghana; however, the rate of unintended pregnancy among this group remains high. We conducted a qualitative study to better understand the context and patterns of ECP use among young unmarried women in Ghana.

Methods: We conducted in-depth interviews with unmarried sexually active women aged 18–24 in Accra, Ghana to explore their perceptions, experiences, and opinions regarding sexual relationships and contraceptive methods, and to examine the factors that influence choice of ECPs. A total of 32 young women participated in the study.

Results: Most participants had used ECPs at least once. Participants described being unable to plan for sexual encounters, and as a result preferred ECPs as a convenient post-coital method. Despite being widely and repeatedly used, women feared the disruptive effects of ECPs on the menstrual cycle and were concerned about long-term side-effects. ECPs were sometimes used as a back-up in cases of perceived failure of traditional methods like withdrawal. Misinformation about which drugs were ECPs, correct dosage, and safe usage were prevalent, and sometimes spread by pharmacists. Myths about pregnancy prevention techniques such as urinating or washing after sex were commonly believed, even among women who regularly used ECPs, and coincided with a misunderstanding about how hormonal contraception works.

Conclusions: ECPs appear to be a popular contraceptive choice among young urban women in Ghana, yet misinformation about their correct usage and safety is widespread. While more research on ECP use among young people is needed, these initial results point to the need to incorporate information about ECPs into adolescent comprehensive sexuality education and youth-friendly services and programmes.

Keywords: Adolescent health, Emergency contraception, Contraception, Reproductive health

Plain English summary

Over the past decade, awareness and use of emergency contraceptive pills (ECPs) among young women has rapidly increased in Ghana. However, the rate of unintended pregnancy among this group remains high.

In this study, we interviewed unmarried women aged 18–24 in Accra, Ghana to understand when, how, and why they use ECPs. A total of 32 women participated in the study.

Most participants had used ECPs at least once. Participants described being unable to plan for sexual encounters, and as a result preferred ECPs because they can be taken after sex. Despite being widely and repeatedly used, women feared that ECPs would cause them harm by disrupting their menstrual cycle and were also concerned about long-term side-effects. ECPs were sometimes used as a back-up in cases when women used a traditional method like withdrawal, but were worried that it had not been effective. Women were also misinformed about which drugs were ECPs, and the correct dosage and safe usage of ECPs. Pharmacists were sometimes the source of the wrong information concerning ECPs. Myths about pregnancy prevention techniques



^{*} Correspondence: sonja.merten@swisstph.ch

²Society, Gender and Health Unit, Department of Epidemiology and Public Health, Swiss Tropical and Public Health Institute, Socinstrasse 57, 4002 Basel, Switzerland

³University of Basel, Petersplatz 1, 4001 Basel, Switzerland Full list of author information is available at the end of the article

such as urinating or washing after sex were commonly believed, even among women who regularly used ECPs.

In conclusion, ECPs appear to be a popular contraceptive choice among young women in Accra, yet misinformation about their correct usage and safety is widespread. While more research on ECP use among young people is needed, these initial results point to the need to incorporate information about ECPs into adolescent comprehensive sexuality education and youth-friendly services and programmes.

Background

Worldwide, 44% of pregnancies are unintended [1]. Unintended pregnancies increase risk of maternal and infant mortality and morbidity, as well as reduce women's prospects for education and employment [1, 2]. In sub-Saharan Africa (SSA), 30–43% of unintended pregnancies end in induced abortion, where 3 out of 4 abortions are classified as unsafe [1, 3].

Emergency contraceptive pills (ECPs) are an essential element of contraceptive services to reduce risk of unintended pregnancy. ECPs are indicated for women who have had unprotected sex, for example due to method failure, incorrect use or non-use of contraceptives, or sexual assault [4]. They have been developed and marketed as an "emergency" or "back-up" method in order to discourage regular use [5]. Over the past decade, knowledge and use of ECPs has increased in many SSA countries, particularly among unmarried women in urban areas [6]. As access to ECPs has increased, studies have found that young women may prefer emergency contraception to long-term methods of contraception, as a convenient post-coital method with fewer side effects [5, 7–9].

In Ghana, 80% of sexually active unmarried women knew about emergency contraception in 2017, up from 49% in 2008 [10, 11]. In 2017, ECPs were the most commonly used method of contraception (either modern or traditional) among sexually active unmarried women aged 20–24 [10]. Other recent studies show similar findings. A 2018 study of youth aged 15–24 in Ghana found that ECPs were the second most popular modern or traditional method among women, next to condoms [12]. A 2016 study of female Ghanaian university students found that 63% had heard of ECPs and 37% had used ECPs [13]. A 2014 study of ECP awareness and use in the northern region of Ghana found that 69% of women aged 15–49 were aware of ECPs; among these, 40% had ever used them [14].

Yet, while use of ECPs has increased among young women, sexual and reproductive health (SRH) indicators have not improved. Results from the 2017 Ghana Maternal Health Study found that nearly half of women aged 20–24 began having sex before age 18, yet less than a third of sexually active unmarried women use a modern

form of contraception [10]. In this context, the rate of unintended pregnancies is high: about 16% of women aged 20–24 have ever had an induced abortion, and a third of births to women aged 20–24 are mistimed or not wanted at all [10, 15]. About 14% of women aged 15–19 have begun childbearing, a proportion that has remained stagnant since 1998 [10].

Given the growing awareness and use of ECPs in Ghana and other countries in SSA, it is important to examine why increased use of ECPs has not translated into improved SRH among young women. The aim of this qualitative study is to better understand the context and patterns of ECP use among young unmarried women in Accra, Ghana. We explore women's perceptions, experiences, and opinions regarding sexual relationships and contraceptive methods, and examine the factors that influence choice of ECPs. Finally, we discuss our results in the context of expanding sexuality education curricula as well as national reproductive health policy guidelines to incorporate information about ECPs.

Methods

Study setting and participants

This study was conducted in April 2016 in Accra, Ghana. The participants were drawn from a sample of participants who completed a randomized controlled trial in Ghana that investigated the effectiveness of sexuality education text messaging programs on knowledge of SRH and on sexual behaviour [16, 17]. Briefly, the original study sent informational text messages about reproductive health to participants for 12 weeks. The inclusion criteria for the original trial was girls aged 14–24 in their second year of secondary school in the Greater Accra region. Participants completed self-administered questionnaires at baseline, midline (3 months later), and endline (15 months later).

The study found that the text messaging programs improved knowledge of reproductive health, but did not have a clear and significant impact on sexual behaviour or use of contraception, compared to the control group. In order to further understand the results of the study, we conducted in-depth interviews one year after the end of the original trial. We used purposive sampling, inviting women who were in the treatment arms of the original trial (i.e. had received sexuality education text messages) and who had reported that they had sex in the past year at endline. We also sampled those who claimed they had ever been pregnant at endline in order to understand the contexts of those pregnancies.

Study design

We conducted in-depth individual interviews following a topic guide. Topics included perceptions about community norms surrounding sex and contraception, personal sexual experiences, attitudes about and experiences with use of modern and traditional methods of contraception, perceptions of barriers to use of contraception, experiences with pregnancy and abortion, experiences and community norms surrounding 'sexting' (exchanging explicit sexual messages via mobile phones), and feedback on the text messaging program they were involved in. The average interview length was 53 min with a range of 32 min to 1 h 34 min.

Interviews were conducted in local language (Twi, Ga, or Ewe) or English based on respondent preference. Two female interviewers of about the same age as respondents were trained on qualitative methods of interviewing in a 2-day training. Training materials were adapted from a field guide created by Family Health International [18]. Interviews were audiotaped, and then translated into English and transcribed simultaneously by a professional Ghanaian translator. To ensure validity, the senior field manager verified a sample of the translations.

We obtained written informed consent from each participant. Participants were informed that participation was voluntary, they could stop the interview at any time, and data would be treated confidentially. The recorded interviews did not include the participant's name. As a token of appreciation, every respondent received 5GHC in airtime phone credit (about 1.30USD). Participants were also provided a free hotline phone number, operated by Marie Stopes International, in case of any further questions or concerns about SRH or their health. Institutional Review Board (IRB) approval was granted by the Committee on the Use of Human Subjects in Research at Harvard University (IRB13-1647) as well as the Ghana Health Service Ethical Review Committee (GHS-ERC:05/09/13). Thirty-two women signed a letter of consent and the final sample included all 32.

Data analysis

The interviews were analyzed using a thematic approach that allowed for themes to emerge through an iterative process of coding and discussion [19]. A preliminary list of codes was created using the topic guide. Transcripts were then independently reviewed by two members of the research team who assigned preliminary codes. The research team members met and discussed their findings, and codes were subsequently added, removed, and refined during these discussions. This process continued until the research team agreed on the final themes and categories. To ensure reliability, all cases of conflicting results were discussed thoroughly in an open process until consensus was reached. This process included comprehensive data treatment and deviant case analysis, in which research team members actively sought out and identified similarities and differences across accounts to ensure different perspectives were represented [20].

Data management and analysis were conducted using the computer software package NVivo™ software Version 11 (QSR International, Doncaster, Victoria, Australia). The study preserved a well-documented audit trail of materials to ensure transparency.

Results

The demographic characteristics of the participants are shown in Table 1. Participants were between 18 and 24 years of age. All 32 participants were unmarried; however 97% were in a relationship. The median age at first sex was 18 years. Six participants had gotten pregnant since becoming sexually active; all six had an induced abortion. None of the participants had children.

Table 1 Demographic characteristics of women participating in in-depth interviews

Characteristic	Total N = 32
Age, years [median (range)]	20 (18–24)
Religion [n (%)]	
Muslim	6 (19)
Catholic	2 (6)
Protestant	8 (25)
Pentecostal/Charismatic	16 (50)
Ethnicity [n (%)]	
Akan	10 (31)
Ga	7 (22)
Ewe	7 (22)
Other/Missing	8 (25)
Mother's education [n (%)]	
Less than secondary	17 (53)
More than secondary	10 (31)
Don't know/Missing	5 (16)
Father's education [n (%)]	
Less than secondary	4 (13)
At least secondary	18 (56)
Don't know/Missing	10 (31)
Marital status [n (%)]	
Married	0
Currently in a relationship	31 (97)
Single	1 (3)
Age at first sex [median (range)]	18 (15–20)
Ever pregnant [n (%)]	6 (19)

Notes: Information on age, marital status, age at first sex, and ever pregnant was obtained from in-depth interviews. Information on religion, ethnicity, and parental education was obtained from baseline data from the original trial from which respondents were recruited [16]

The most common forms of contraception that participants had tried at least once were condoms and ECPs, followed by withdrawal and using a menstrual calendar. Only 5 participants mentioned using oral contraceptive pills (OCPs) and only 1 had used an injectable. The most common ECP brand was Postinor-2° (Levonorgestrel); others that were mentioned were Lydia Postpil° (Levonorgestrel), Lenor° (Levonorgestrel), and (incorrectly) Primolut-N° (Noresthiesterone). Primolut-N is misused off-label as an ECP or a pre-coital contraceptive in Ghana [4, 21, 22].

Themes arising from the interviews are shown in Table 2. The major themes that emerged were "unwanted and unplanned context of sexual encounters", "popularity of ECPs", "ECPs as back-up for traditional methods", "negative perceptions of side effects of ECPs", "misinformation about ECPs", "negative perceptions of condoms", and "misunderstanding of human reproduction and contraception".

Unwanted and unplanned context of sexual encounters

Participants described sexual encounters as largely unplanned for a variety of reasons. Coerced or pressured sexual encounters were mentioned by the majority of participants, either as something they experienced themselves or something they noticed in the community. While six participants described having been physically forced into sexual activity, most described these encounters as partners putting pressure on them to have sex over time. Participants described their first time having sex as necessary to prove their love to their partner or else risk dissolution of the relationship.

"Well, to me, he was like, we should try it. And I said, no, I wasn't ready for that. He was like, "for the first time then..." – as for him he thinks I don't love him because we've dated for 2 years without sex." (Age 21)

Many participants also described sex as an obligation in return for financial benefits. Participants described feeling indebted to boyfriends who provide schooling fees or take care of them financially, and feeling that eventually they had to repay the boyfriend with sex. One participant said,

Table 2 Themes arising from in-depth interviews

Theme	Categories
Unwanted and unplanned context of sexual encounters	Coerced or pressured sex
	Sex in return for financial benefits
	Impromptu sex
Popularity of ECPs	Post-coital method
	Diffusion of information about it
	Can take without partner's knowledge
	ECPs easier to take than OCPs because fewer pills
ECPs as back-up for traditional methods	Lack of understanding of calendar method
	Lack of trust in partners to practice withdrawal correctly
	ECP can be used as backup in case of failure
Negative perception of side effects of ECPs	Dislike/fear of change in menses
	Fear/lack of clarity of side effects
	Fear of "addiction" / pills will become ineffective
Misinformation about ECPs	Lack of clarity as to which drugs are ECPs vs OCPs
	Misunderstanding about how to take ECPs
	Pharmacist providing incorrect information
	Difficulty in obtaining ECPs due to cost, shyness, or pharmacist refusal to sell
Negative perception of condoms	Negative representation in society (representing promiscuity or infidelity)
	Lack of trust in condoms
	Reduced sexual pleasure
Misunderstanding of human reproduction and contraception	Pregnancy prevention myths
	Lack of understanding about how contraceptives work
	Lack of timely sexual health education or parental support/communication

"Because of the hardship I was going through that was why I went in for a guy, so if he provides my basic needs then if he ask me for sex and maybe I don't give him, I will think he will go for another girl, which means I will forgo my basic needs." (Age 22)

Another participant described a common situation in her neighbourhood:

"Girls may need something like pads, when you ask your mother and she doesn't get it for you, you will definitely go and ask a guy. He will give it to you but not for free. It's only a few that will do that for free. When he gives you today, tomorrow he will demand he have sex with you first." (Age 18)

Finally, participants described sexual encounters, even wanted, as unexpected. Unplanned encounters happened with both regular partners and via casual sex. For example, a participant described her inability to plan for sex:

"I think sorry to say, you don't prepare to have sex, that's why I did it. We don't prepare to have sex. You can say you're not going to have sex with your man or your woman or your wife; but it is feelings...Yes, you can just have sex without knowing. You know you are doing something but you can't control it, because it's sex. Yes. So, that's what happened." (Age 20)

Previous research has found that when sexual encounters are unplanned, particularly with an imbalanced gender dynamic, young women are less able to control their sexual experiences including the use of condoms and other preventative forms of contraception [23, 24].

Popularity of ECPs

Twenty-six of the 32 participants had used ECPs at least once. The most common brand was Postinor-2. Participants felt that ECPs were easy to take since they could be taken after having sex, rather than planning for it. Others said that aside from condoms, Postinor-2 was the only type of contraception (outside of condoms) they knew about. A number of participants mentioned hearing recommendations about ECPs from friends or aunties. Upon being asked why a participant chose to use Postinor-2, she replied,

"That was the only one I know of... I asked my friend and she told me that one was very good so I should try it. So when I tried it I realized it was good." (Age 20)

Additionally, some participants preferred ECPs so that they could take it without their partner's knowledge. Some participants described how a partner wanted them to become pregnant, even though they were not ready, so they took ECPs without informing their partners. One participant described her situation:

"For him he said he wants a child but I told him I'm not ready yet so if I tell him to do [use a condom], he wouldn't do it but on my side, I prevent myself [from getting pregnant]." (Age 22)

Finally, some participants chose to use ECPs over OCPs because they felt they could not trust themselves to remember to take a pill each day. One participant said,

"I chose Postinor-2 because the other ones I know; something like Secure [a brand of OCP] they said you will take it every day. And when you missed one day you can get pregnant. So I prefer Postinor-2 after having sex within 72 hours you have to take it. So after me having sex then I take it and then I would be free. I prefer that than Secure because I may not get the time to take the medicine everyday." (Age 19)

ECPs as back-up for traditional methods

While the majority of participants had attempted traditional calendar or withdrawal methods, they also described being fearful of the inefficacy of these methods. Many did not trust men to practice withdrawal correctly, and were afraid of becoming pregnant after its use. Some participants discussed taking ECPs after a perceived failure of withdrawal; for example, one 20-year-old participant had the following exchange with the interviewer:

Respondent: I think when the guy is about to release, the lady realizes but the guy because of the fantasies it doesn't occur to him then he will release inside and pretend as if he didn't.

Interviewer: So using this method that you are saying you have used before were you scared of getting pregnant?

Respondent: I was really scared.

Interviewer: So what did you do?

Respondent: It was after that I went to buy Postinor-2 and took it just to prevent anything.

Similarly, for the calendar method, many participants described not understanding how to calculate their safe period, particularly when their menstrual cycle changes. At the same time, participants were eager to understand these methods and felt that they had not been provided any information on them.

Negative perceptions of side effects of ECPs

Fears, dislikes, and misunderstanding of hormonal methods (both OCPs and ECPs) were mentioned by most participants. Participants mentioned both a dislike and a fear of any change in the schedule or heaviness of flow of their menses. Participants admitted that they were not sure about side effects of hormonal methods, and during the course of the interview, a few asked about the side effects of ECPs in particular. A number of participants mentioned fear of hormonal pills, including ECPs, as damaging the womb and preventing future ability to conceive. One participant explained,

"Definitely, everything has its disadvantage. I have learnt that taking too much of Postinor-2, it will come to a time you can't give birth." (Age 18)

Other participants believed that taking too many pills would lead the participant to become "addicted" to them in such a way as for the contraception to become ineffective at preventing pregnancy. One participant discussed why she stopped taking Postinor-2:

"Because if you continually take the tablets or the pills, it will get to a point where your body becomes used to; so it would not work... And also I know that it will get to a point where I will get complications with it because in everything there is a bad side and there is a good side. So I can say I'm using [Postinor-2] to prevent pregnancy but maybe I may develop some infections from it. So I've stopped using it." (Age 19)

While both ECPs and OCPs elicited similar reactions, ECPs appeared to be more tolerated, most likely due to the low number of pills (usually two) and therefore the relatively short-term effect they had on the menstrual cycle as compared to OCPs. Participants did not mention dosage at all; instead taking "too much" medication referred to taking too many individual pills. One respondent mentioned that she had thought about taking OCPs but was concerned about how taking a pill every day would impact her health. The idea that taking a medicine occasionally rather than daily would imply that the health impact is less has been found in other studies of ECPs in West Africa [25].

Misinformation about ECPs

Misinformation about how to safely use ECPs was prevalent among participants, some of which was spread by pharmacists. For example, one participant described why she stopped using Postinor-2:

"Where I've been buying [Postinor-2], the pharmacist advised me that Postinor-2 must be taken [just] once in a year. That's what he told me." (Age 22)

Some participants were confused as to how to take ECPs, for example, they described taking both pills of Postinor-2 at the same time (whereas the correct dosage is one pill as soon as possible after unprotected sex, and the second pill 12 hours later). Pharmacists also misinformed participants about how to correctly take ECPs. There was also confusion among participants as to which drugs were ECPs. As previously mentioned, Primolut-N, which is indicated for treatment of endometriosis, menorrhagia and dysmenorrhea, is used off-label both as a pre-coital method and an ECP in Ghana, though the effectiveness has not been studied and no health organization or agency has recommended its use [4, 21, 22].

About half of participants mentioned difficulty in buying ECPs at drug shops due to their own shyness, the cost, or the pharmacist's refusal to sell it, while the rest thought it was easy to acquire.

Negative perceptions of condoms

Participants had largely negative perceptions and experiences with condoms, a finding consistent with previous qualitative literature [26]. First, many participants felt that condoms represent distrust and lack of love and commitment in a relationship. One participant said about asking a partner to use a condom:

"Some would say maybe if you use — maybe if you tell the man, he will say you are not that into me that is why you want us to use the condom, that's what I really know. Somebody will say you don't love me that's why you are saying or maybe you think I have a sickness that's why you want us to use a condom." (Age 22)

Use of condoms also represented promiscuity on the part of women; participants said that if a male partner wanted to use a condom, it was because he thought the woman was cheating or, in one participant's opinion, that he thought of the woman as a prostitute.

Additionally, a number of participants mentioned a lack of trust in the effectiveness of condoms. Participants feared that condoms may easily tear and burst. Some participants were fearful of the condom tearing without their knowledge and resulting in unintended pregnancy;

one participant mentioned that a condom bursting would be harmful to the womb.

Finally, 23 of the 32 participants mentioned that condoms reduce the pleasure of sex, either as their own opinion, their partner's opinion, or as community norms. Many participants mentioned the analogy of using condoms to "eating toffee in a wrapper". Similar findings were found among studies with youth in Uganda [27], South Africa [28], and Nigeria [29], and among married women in Malawi [30].

Misunderstanding of human reproduction and contraception

Participants in general had a considerable misunderstanding of the biological process of how women become pregnant, leading to a wide variety of pregnancy prevention myths. Common myths included drinking a large amount of water or a sugary beverage to "urinate out" sperm, standing up quickly after sex, using a finger to pull out the sperm, and taking paracetamol. Similar myths have been documented in previous studies in both Ghana and other countries [5, 31, 32]. The myths also coincided with misunderstandings about how hormonal contraception works. For example, one 18-year-old participant described using ECPs regularly in order to prevent pregnancy. When asked how ECPs worked, she said,

"After sex when you take those ones it washes away the sperms from your system. It comes in a form of urine."

Then she went on to describe preventing pregnancy when she does not use contraception:

"But sometimes, it's when I don't take the [EC] pill. I drink a lot of water after sex. ... It also helps me urinate [the sperm out]"

Other participants described similar misunderstandings regarding how ECPs work, and in general, the way that hormonal contraceptives work to prevent a pregnancy was not known to most participants. Most participants felt that they had not received enough education about SRH. At the time they began having sex, many participants said they were in Junior High School rather than Senior High School, and that therefore their education on sexual health was inadequate. While average age at first sex is 18 years for girls in Ghana, it is common for girls around this age to be attending Junior High School due to late entry to school, repetition, and drop out and drop in [33].

Discussion

The results of this study provide evidence on how the context of sexual encounters and the perceptions and

experiences of contraception among young sexually active unmarried women in Accra, Ghana influence the use of ECPs. A number of important conclusions emerged. First, sexual encounters are often unwanted or unplanned. Young women rely on their partners financially for school fees and personal items, providing sex in return. This may reduce women's control over the use of condoms and preventative methods of birth control, and drive women towards the use of ECPs which are within their sphere of control. A review of 45 quantitative and qualitative studies in SSA found that adolescent girls engage in sexual relationships with older men for financial benefits; while girls have, to a great extent, control over choice of partners, duration of relationships, and the start of sexual relations, they have little control over sexual practices within partnerships including condom use and violence [24]. Second, despite the widespread use of ECPs among participants, ECPs are nevertheless mistrusted due to the disruptive effect they have on the menstrual cycle. There is a pervasive belief that taking too many hormonal pills will have a negative effect on future fertility; there is also a belief that ECPs will become ineffective at preventing pregnancy if taken too many times. These findings are consistent with other qualitative studies [7, 34, 35]. Third, traditional methods such as withdrawal and following a menstrual calendar are used; however, when there is a perception of method failure, ECPs are sometimes used as a back-up method. Finally, there is a great deal of misinformation about ECPs, including which drugs are ECPs, how to take ECPs, and how frequently ECPs can be safely taken, some of which was spread by pharmacists.

We found that a large majority of participants had experience using ECPs; participants liked that ECPs could be used post-coitally and without their partner's knowledge. While the rate of use may be higher in our study because participants were sent an informational message about ECPs during the course of the original trial, many participants mentioned hearing about ECPs from friends and family. We also found that there is a substantial misunderstanding of the biology of human reproduction, which leads to myths about nonsensical pregnancy prevention techniques such as urinating out sperm. These myths are widespread and are believed even among women who regularly use modern contraception. A study of Nigerian University students also found that women combine ECPs with saline solution douches or with other medications to increase their certainty of preventing a pregnancy [34, 35]. Interestingly, participants in our study had received information on the ineffectiveness of some prevention techniques, including bathing after sex and standing up during sex, during the original trial [16, 17]. While the messaging intervention was effective at improving overall SRH knowledge, it appears that these messages were not comprehensive enough to

dispel the widespread myths concerning pregnancy prevention.

Our study contributes to the larger literature on ECPs in SSA. Recent studies have found that levels of awareness and use of ECPs among young women vary considerably across countries in SSA [8, 36–39]. Consistent with the findings of this study, previous research has found that young women generally find ECPs an acceptable and convenient option that fill the need of a post-coital contraception that can be taken as needed [4, 5, 7, 25]. However, our results also contribute to the growing evidence on barriers to ECP access and use, including myths and misconceptions about risks of ECPs [5, 25, 34, 35, 40], poor knowledge of correct usage [13, 34, 39], and restrictive distribution practices of pharmacists and health care providers [41–44].

A critical step forward to reducing misinformation and misuse of ECPs would be to incorporate information on ECPs into comprehensive sexuality education (CSE) in schools. Ghana adopted an Adolescent Reproductive Health Policy in 2000, which led to the inclusion of a reproductive health component in the educational curriculum at primary, junior high, and senior high levels; however, a 2017 report on implementation of SRH education policies in Ghana found that fewer than half of the students reported that they had learned about contraceptive methods [45]. The majority of students who started learning SRH in senior high school said that they would have liked to have started earlier, a finding that is supported in our study. Among teachers, only 54% reported teaching about ECPs at all, while 99% reported teaching about condoms and 83% reported teaching about OCPs [45]. Educational curricula should include a wider range of contraceptive options, along with information on the mechanisms of how methods work and a comprehensive, accurate discussion of their side effects. In addition, our findings provide more evidence that gender and power need to be addressed within CSE, including topics of partner communication, equitable relationships, and harmful gender norms [46, 47]. Finally, as young women demonstrate high rates of pharmacy utilization for obtaining contraceptives [48], pharmacists and health care providers also need comprehensive training on correct dosage, safe usage, and side effects of ECPs. Previous research has found low knowledge of ECPs among providers in Ghana [41] and Senegal [49].

Lastly, reproductive health policies need to address the growing access to and use of ECPs among youth. The Ghana Health Service Family Health Division 2015 Annual Report, for example, details a Family Planning Programme describing trends and progress in uptake of long-acting contraceptive methods, OCPs, injectables, male and female condoms, and permanent methods, but makes no mention of ECPs at all [50]. Moreover, guidelines that address ECPs

should take the social and sexual context of young women's sexual experiences as well as their preferences into account. While ECPs were originally developed as an "emergency" method, our results support the findings of other studies that young women prefer ECPs as a post-coital method and use it repeatedly [5, 7, 8]. While ECPs are not as effective as preventative methods of contraception, repeated use has been found to be safe, though more evidence is needed [51]. Results from our study show that women are reluctant to use condoms and OCPs consistently, so the idea of "bridging" young women who use ECPs to long-term preventative methods of contraception may be challenging, a finding consistent with previous literature [4, 5]. We also find that young women are fearful of overuse of ECPs and are worried about their side effects. Therefore, misinformation about the dangers of repeated use and side effects of ECPs may lead women to rely on ineffective and possibly dangerous "prevention" methods (like urinating or taking a large dose of paracetamol). Conversely, encouraging young women to use ECPs regularly may increase the spread of sexually transmitted infections (STIs). Overall, more research is needed to understand how uptake of ECPs affects rates of STIs.

Strengths and limitations

Our study has strengths and limitations. First, as this is a qualitative study, our results are not generalizable to other contexts and populations beyond young unmarried women in Accra, Ghana. In addition, participants in our study were originally recruited as secondary school students in Accra, thus our results also may not generalize to other demographic or geographic groups in Ghana. However, we find that characteristics of our sample, such as median age at first sex, are similar to those of women aged 20-24 in Ghana [15]. In addition, many of our findings are consistent with those of other studies conducted in Ghana [5, 7, 13]. Second, participants received a 12-week sexuality education intervention as part of the original trial, which may have impacted on their behaviour and thus their sexual experiences and preferences. However, this is a strength of our study as well in that interviewer debrief notes indicated that the fact that participants had received information from the sexuality education program (with which the interviewer was affiliated) allowed participants to feel more comfortable with sharing personal sexual information with the interviewer. For example, six participants provided long and detailed stories about their abortion experiences, indicating that they trusted the interviewers to discuss this information. Since sex is often a taboo topic in Ghana, interviewer trust is essential to obtaining complete and honest responses and ensuring credibility of the study.

Conclusions

Continued investment in adolescent CSE, youth-friendly health services, and effective youth and community programs are essential to protect sexual and reproductive health and rights [52–54]. However, for programs to be effective they must be tailored to the particular needs of sexually active youth. Our results show that ECPs are an important component of a comprehensive SRH strategy that is currently left out of policy and educational guidelines. Young women's preference for ECPs may be at odds with HIV prevention approaches that solely promote condoms. The development of a youth SRH strategy must consider the social and sexual context of ECP use among young women, as well as how best to integrate ECPs into a dual-protection strategy to reduce risk of unintended pregnancy as well as HIV/AIDS and STIs.

Abbreviations

CSE: Comprehensive sexuality education; ECP: Emergency contraceptive pill; IRB: Institutional Review Board; OCP: Oral contraceptive pill; SRH: Sexual and reproductive health; SSA: Sub-Saharan Africa; STI: Sexually transmitted infection

Acknowledgements

We thank Shannon Glaspy for research support and Günther Fink for support of the study. We are grateful to the participants who took the time to share their experiences with us. We thank Comfort Bonney Arku, Grace Gletsu, Charles Sefenu, Christine Papai, and the IPA staff who contributed to this study for their hard work and support. We thank Grace Kafui Annan and staff at the Ghana Health Service Health Promotion Unit for their guidance and support.

Funding

This research was funded by the Weiss Family Fund for Research in Development Economics, the Harvard Lab for Economic Applications and Policy, and the Harvard Institute for Quantitative Social Science. Study sponsors had no role in study design, data collection, analysis, interpretation, writing of the report, or decision for publication.

Availability of data and materials

The data used during the current study are available from the corresponding author on reasonable request.

Authors' contributions

SR analyzed and interpreted the data. All authors edited the manuscript. All authors read and approved the final manuscript.

Ethics approval and consent to participate

Institutional Review Board (IRB) approval was granted by the Committee on the Use of Human Subjects in Research at Harvard University (IRB13–1647) as well as the Ghana Health Service Ethical Review Committee (GHS-ERC:05/09/13). We obtained written informed consent from each participant.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Geary Institute, University College Dublin, Dublin, Ireland. ²Society, Gender and Health Unit, Department of Epidemiology and Public Health, Swiss

Tropical and Public Health Institute, Socinstrasse 57, 4002 Basel, Switzerland. ³University of Basel, Petersplatz 1, 4001 Basel, Switzerland.

Received: 20 September 2018 Accepted: 5 December 2018 Published online: 19 December 2018

References

- Bearak J, Popinchalk A, Alkema L, Sedgh G. Global, regional, and subregional trends in unintended pregnancy and its outcomes from 1990 to 2014: estimates from a Bayesian hierarchical model. Lancet Glob Health. 2018;6: e380–9.
- Singh S, Darrroch JE, Ashford LS. Adding it up: the costs and benefits of investing in sexual and reproductive health [internet]. New York: Guttmacher Institute and United Nations Population Fund; 2014. Available from: https://www.unfpa.org/sites/default/files/pub-pdf/Adding%20lt%20Up-Final-11.18.14.pdf
- Ganatra B, Gerdts C, Rossier C, Johnson BR, Tunçalp Ö, Assifi A, et al. Global, regional, and subregional classification of abortions by safety, 2010–14: estimates from a Bayesian hierarchical model. Lancet. 2017;390:2372–81.
- Chin-Quee D, Hinson L, L'Engle KL, Otterness C, Janowitz B. Bridge over troubled waters: considerations in transitioning emergency contraceptive users to hormonal methods. Contraception. 2012;85:363–8.
- L'Engle KL, Hinson L, Chin-Quee D. "I love my ECPs": challenges to bridging emergency contraceptive users to more effective contraceptive methods in Ghana. J Fam Plann Reprod Health Care. 2011;37:146–51.
- Morgan G, Keesbury J, Speizer I. Characteristics and patterns of use of emergency contraception among urban women in Nigeria and Kenya. Stud Fam Plan. 2014;45:59–72.
- Both R. Young People's use and perceptions of emergency contraceptives in sub-Saharan Africa: existing insights and Knowledge gaps. Sociol Compass. 2013;7:751–61.
- Both R. Emergency contraceptive use in Addis Ababa, Ethiopia: challenging common assumptions about young people's contraceptive practices. Reprod Health Matters. 2015;23:58–67.
- Cover JK, Drake JK, Kyamwanga IT, Turyakira E, Dargan T, Kumakech E, et al. Consumer perspectives on a Pericoital contraceptive pill in India and Uganda. Int Perspect Sex Reprod Health. 2013;39:195–204.
- Ghana Statistical Service (GSS), Ghana Health Service (GHS), and ICF. Ghana Maternal Health Study 2017 [Internet]. Accra, Ghana: GSS, GHS, and ICF; 2018. Available from: https://dhsprogram.com/pubs/pdf/FR340/FR340.pdf
- Ghana Statistical Service (GSS), Ghana Health Service (GHS), and ICF Macro. Ghana Demographic and Health Survey 2008 [Internet]. Calverton, MD: ICF Macro; 2009. Available from: https://dhsprogram.com/pubs/pdf/FR221/ FR221[13Aug2012].pdf
- Grindlay K, Dako-Gyeke P, Ngo TD, Eva G, Gobah L, Reiger ST, et al. Contraceptive use and unintended pregnancy among young women and men in Accra, Ghana. PLoS One. 2018;13:e0201663.
- Darteh EKM, Doku DT. Knowledge and usage of emergency contraceptives Among University students in Ghana. J Community Health. 2016;41:15–21.
- Amalba A, Mogre V, Appiah MN, Mumuni WA. Awareness, use and associated factors of emergency contraceptive pills among women of reproductive age (15-49 years) in tamale, Ghana. BMC Womens Health. 2014;14:114.
- Ghana Statistical Service (GSS), Ghana Health Service (GHS), and ICF International. Ghana Demographic and Health Survey 2014 [Internet]. Rockville, Maryland: GSS, GHS, and ICF International; 2015. Available from: http://dhsprogram.com/pubs/pdf/FR307/FR307.pdf
- Rokicki S, Cohen J, Salomon JA, Fink G. Impact of a text-messaging program on adolescent reproductive health: a cluster–randomized trial in Ghana. Am J Public Health. 2016;107:298–305.
- Rokicki S, Fink G. Assessing the reach and effectiveness of mHealth: evidence from a reproductive health program for adolescent girls in Ghana. BMC Public Health. 2017;17:969.
- Mack N, Woodsong C, MacQueen KM, Guest G, Namey E. Qualitative Research Methods: A data collector's field guide [Internet]. Family Health International 360; 2005. Available from: https://www.fhi360.org/resource/ qualitative-research-methods-data-collectors-field-guide.
- Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3:77–101.
- Silverman D. Doing qualitative research: a practical handbook. Thousand Oaks: SAGE Publications Limited; 2013.

- 21. Opare-Addo HS, Britwum PK, GAO A. A Study of the use of Primolut N tablet as a contraceptive in the Kumasi Metropolis of Ghana. Afr J Reprod Health. 2011;15:65–7.
- Krakowiak-Redd D, Ansong D, Otupiri E, Tran S, Klanderud D, Boakye I, et al. Family planning in a sub-district near Kumasi, Ghana: side effect fears, unintended pregnancies and misuse of a medication as emergency contraception. Afr J Reprod Health. 2011;15:135–47.
- Elouard Y, Weiss C, Martin-Hilber A, Merten S. Sexual violence as a risk factor for family planning-related outcomes among young Burundian women. Int J Public Health. 2018;63:13–22.
- Luke N. Age and economic asymmetries in the sexual relationships of adolescent girls in sub-Saharan Africa. Stud Fam Plan. 2003;34:67–86.
- Teixeira M, Guillaume A, Ferrand M, Adjamabgo A, Bajos N. Representations and uses of emergency contraception in West Africa. A social anthropological reading of a northern medicinal product. Soc Sci Med. 2012;75:148–55
- Williamson LM, Parkes A, Wight D, Petticrew M, Hart GJ. Limits to modern contraceptive use among young women in developing countries: a systematic review of qualitative research. Reprod Health. 2009;6:3.
- Nalwadda G, Mirembe F, Byamugisha J, Faxelid E. Persistent high fertility in Uganda: young people recount obstacles and enabling factors to use of contraceptives. BMC Public Health. 2010;10:530.
- 28. MacPhail C, Campbell C. 1 think condoms are good but, aai, I hate those things': condom use among adolescents and young people in a Southern African township. Soc Sci Med. 2001;52:1613–27.
- Sunmola AM. Developing a scale for measuring the barriers to condom use in Nigeria. Bull World Health Organ. 2001;79:926–32.
- John NA, Babalola S, Chipeta E. Sexual pleasure, partner dynamics and contraceptive use in Malawi. Int Perspect Sex Reprod Health. 2015;41:99–107.
- Widyastari DA, Shaluhiyah Z, Widjanarko B. Urinating after sexual intercourse prevents pregnancy: adolescents' misconceptions of reproductive health Knowledge. J Kesehat Reproduksi. 2011;1:102–12.
- Xu J, Cheng L. Awareness and usage of emergency contraception among teenagers seeking abortion: a Shanghai survey. Eur J Obstet Gynecol Reprod Biol. 2008;141:143–6.
- Associates for Change, Ghana. The Ghana Girls' Education Mapping Study (2000-2011) Final Report [Internet]. Accra, Ghana: CAMFED, Ghana; 2011. Available from: http://web.net/~afc/download3/Education%20Research/Camfed%20Girls%20 Education%20Study/Girls%20Education%20Mapping%20Report%20Review% 20Oct%2031%20final.pdf
- Ajayi Al, Nwokocha EE, Akpan W, Adeniyi OV. Use of non-emergency contraceptive pills and concoctions as emergency contraception among Nigerian University students: results of a qualitative study. BMC Public Health. 2016;16:1046.
- 35. Ajayi Al, Nwokocha EE, Adeniyi OV, Ter Goon D, Akpan W. Unplanned pregnancy-risks and use of emergency contraception: a survey of two Nigerian Universities. BMC Health Serv Res. 2017;17:382.
- Palermo T, Bleck J, Westley E. Knowledge and use of emergency contraception: a multicountry analysis. Int Perspect Sex Reprod Health. 2014; 40:079–86
- 37. Westley E, Kapp N, Palermo T, Bleck J. A review of global access to emergency contraception. Int J Gynecol Obstet. 2013;123:4–6.
- Koyama A, Hagopian L, Linden J. Emerging options for emergency contraception. Clin Med Insights Reprod Health. 2013;7:CMRH.S8145
- Ezebialu IU, Knowledge EAC. Practice of emergency contraception among female undergraduates in south eastern Nigeria. Ann Med Health Sci Res. 2014;3:541–5.
- Westley E, Glasier A. Emergency contraception: dispelling the myths and misperceptions. Bull World Health Organ. 2010;88:243–243A.
- 41. Creanga AA, Schwandt HM, Danso KA, Tsui AO. Knowledge about emergency contraception among family-planning providers in urban Ghana. Int J Gynecol Obstet. 2011;114:64–8.
- Maharaj P, Rogan M. Missing opportunities for preventing unwanted pregnancy: a qualitative study of emergency contraception. BMJ Sex Reprod Health. 2011;37:89–96.
- Both R, Samuel F. Keeping silent about emergency contraceptives in Addis Ababa: a qualitative study among young people, service providers, and key stakeholders. BMC Womens Health. 2014;14:134.
- 44. Dawson A, Tran N-T, Westley E, Mangiaterra V, Festin M. Workforce interventions to improve access to emergency contraception pills: a systematic review of current evidence in low- and middle-income countries

- and recommendations for improving performance. BMC Health Serv Res. 2015;15:180.
- 45. Awusabo-Asare K, Stillman M, Keogh S, Doku DT, Kumi-Kyereme A, Esia-Donkoh K, et al. From Paper to Practice: Sexuality Education Policies and Their Implementation in Ghana [Internet]. New York: Guttmacher Institute; 2017. Available from: https://www.guttmacher.org/report/sexuality-education-ghana
- Haberland NA. The case for addressing gender and power in sexuality and HIV education: a comprehensive review of evaluation studies. Int Perspect Sex Reprod Health. 2015;41:031–42.
- Karim AM, Magnani RJ, Morgan GT, Bond KC. Reproductive health risk and protective factors among unmarried youth in Ghana. Int Fam Plan Perspect. 2003:29:14–24.
- Chandra-Mouli V, Parameshwar PS, Parry M, Lane C, Hainsworth G, Wong S, et al. A never-before opportunity to strengthen investment and action on adolescent contraception, and what we must do to make full use of it. Reprod Health. 2017;14:85.
- Mané B, Brady M, Ramarao S, Thiam A. Emergency contraception in Senegal: challenges and opportunities. Eur J Contracept Reprod Health Care. 2015;20:64–73.
- Ghana Health Service Family Health Division. 2015 Annual report [internet].
 Accra, Ghana: Ghana health service; 2015. Available from: https://www.ghanahealthservice.org/downloads/2015_FAMILY_HEALTH_DIVISION_ANNUAL_REPORT.pdf
- Halpern V, Raymond EG, Lopez LM. Repeated use of pre-and postcoital hormonal contraception for prevention of pregnancy. Cochrane Database Syst Rev. 2010;1:CD007595.
- Chandra-Mouli V, Lane C, Wong S. What does not work in adolescent sexual and reproductive health: a review of evidence on interventions commonly accepted as best practices. Glob Health Sci Pract. 2015;3:333–40.
- Santhya KG, Jejeebhoy SJ. Sexual and reproductive health and rights of adolescent girls: evidence from low- and middle-income countries. Glob Public Health. 2015;10:189–221.
- Chandra-Mouli V, McCarraher DR, Phillips SJ, Williamson NE, Hainsworth G. Contraception for adolescents in low and middle income countries: needs, barriers, and access. Reprod Health. 2014;11:1.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

